Author's response to reviews

Title: The influence of calcium and magnesium in drinking water and diet on cardiovascular risk factors in individuals living in hard and soft water areas with differences in cardiovascular mortality.

Authors:

Dr Christina Nerbrand (christina.nerbrand@skane.se)
Lars Agreus (lars.agreus@klinvet.ki.se)
Ragnhild A Lenner (ragnhild.lenner@nutrition.gu.se)
Per Nyberg (per.nyberg@skane.se)
Kurt Svardsudd (kurt.svardsudd@pubcare.uu.se)

Version: 5 Date: 10 Jun 2003

Competing interests are included in the manuscript.
The figure has been adjusted according to suggestions